Author's response to reviews

Title: Health disparities and advertising content of women's magazines: a cross-sectional study

Authors:

Susan C Duerksen (sduerksen@projects.sdsu.edu)
Amy Mikail (amikail@ucsd.edu)
Laura Tom (farbeyondthestars@hotmail.com)
Annie Patton (annepatton85@hotmail.com)
Janina Lopez (janinalopez@hotmail.com)
Xavier Amador (xamador@ucsd.edu)
Reynaldo Vargas (rvargas@ucsd.edu)
Maria Victorio (mvictori@ucsd.edu)
Brenda Kustin (bkustin@ucsd.edu)
Georgia R Sadler (gsadler@ucsd.edu)

Version: 9 Date: 1 August 2005

Author's response to reviews: see over
Dear BioMed Central Editorial Board:

We are pleased to submit the enclosed revised manuscript (MS: 2088259646487859 - Health Disparities and Advertising Content of Women's Magazines: a cross-sectional study).

The following formatting changes have been made as requested:

1) Word count was removed from the title page.

2) Page breaks were removed, except at the end of the title page.

3) References in the text were placed before the full stop.

4) The figures were cropped to minimize white space around the images. Using Microsoft Word, I was unable to crop the white space above and below the images any more closely, even with the margins set at 0.

5) References in the text already were numbered consecutively and in square brackets.

We will be happy to answer any further questions or make any further changes as needed. Thank you very much for your consideration of this paper.

Sincerely,

Susan Duerksen, MPH
Georgia Robins Sadler, MBA, Ph.D.